Author's response to reviews

Title: Associations of Complement Factor B and Complement component 2 Genotypes with Subtypes of Polypoidal Choroidal Vasculopathy

Authors:

Koji Tanaka (tanaka.koji@nihon-u.ac.jp)
Tomohiro Nakayama (nakayama.tomohiro@nihon-u.ac.jp)
Ryusaburou Mori (ryu-m@sa2.so-net.ne.jp)
Naoyuki Sato (n-satou@hsri-ei.co.jp)
Akiyuki Kawamura (kw-eye-c@xb3.so-net.ne.jp)
Mitsuko Yuzawa (yuzawa.mitsuko@nihon-u.ac.jp)

Version: 7 Date: 5 June 2014

Author's response to reviews: see over
June 5, 2014

Editor-in-Chief
BMC Ophthalmology

Dear Editor,

Enclosed is the revised version of our manuscript entitled “Associations of Complement Factor B and Complement component 2 Genotypes with Subtypes of Polypoidal Choroidal Vasculopathy” (reference ID number 6905275981236409R5), which we are submitting for possible publication in BMC Ophthalmology.

We appreciate the review of our previous submission of this manuscript and the helpful comments sent on May 28, 2014. We have revised the manuscript based on the editorial request.

My coauthors and I are honor to be published in BMC Ophthalmology.

1. Please remove the figure legends from the figures and include these after the references section under the heading: ‘figure legends’.

   →We have revised the manuscript as removing the figure legends from the figures and put the figures into upload file. We put “figure legends” section after the references.

Sincerely yours,

Tomohiro Nakayama, M.D.
Division of Laboratory Medicine, Department of Pathology and Microbiology
Nihon University School of Medicine
30-1 Ooyaguchi-kamimachi, Itabashi-ku, Tokyo 173-8610, Japan
Tel.: +81-3-3972-8111 (Ext. 8205); Fax: +81-3-5375-8076
E-mail: nakayama.tomohiro@nihon-u.ac.jp